Author's response to reviews

Title: Screening Mammography Beliefs and Recommendations: A Web-based Survey of Primary Care Physicians

Authors:

Shagufta Yasmeen (shagufta.yasmeen@ucdmc.ucdavis.edu)
Patrick S Romano (psromano@ucdavis.edu)
Daniel J Tancredi (ditancredi@ucdavis.edu)
Naomi H Saito (nhsaito@ucdavis.edu)
Julie Rainwater (julie.rainwater@ucdmc.ucdavis.edu)
Richard L Kravitz (rlkravitz@ucdavis.edu)

Version: 2 Date: 17 August 2011

Author's response to reviews: see over
August 15, 2011

The Editor
BMC Health Services Research

Dear Editor

Thank you for reviewing our manuscript titled “Screening Mammography Beliefs and Recommendations: A Web-based Survey of Primary Care Physicians”. We have addressed reviewer’s comments and revised the manuscript. Thank you for extending my resubmission deadline. I appreciate it very much.

Please see the attached documents.

Response to reviewer’s comments (Reviewer 1 and 2)
Revised manuscript, tables and figures
Appendix 1. Table attached for additional information for reviewers.

Correspondence and any other communication regarding this manuscript should be directed to me at the following address and phone number:

Sincerely
Shagufta Yasmeen, MD
Department of Obstetrics and Gynecology
University of California, Davis
4860 Y Street, Suite 2500
Sacramento, CA 95817
(916) 734-6557 Fax (916) 734-6666
Sincerely,

Shagufta Yasmeen, MD
Associate Professor
Department of Obstetrics/Gynecology and General Internal Medicine
University of California Davis School of Medicine